Author's response to reviews

Title: School difficulties and role of social, material, behavioural, physical and mental resources among multi-cultural students

Authors:

Kénora Chau (c.kenora@yahoo.fr)
Michèle Baumann (michele.baumann@uni.lu)
Bernard Kabuth (kabuth_b@yahoo.fr)
Nearkasen Chau (Nearkasen.Chau@wanadoo.fr)

Version: 3 Date: 14 January 2012

Author's response to reviews: see over
Concerne: revised article MS: 1089038875657928

Dear Editor,

We would like to submit for reviewing and, hopefully, publication in your Journal, a manuscript for research entitled:

**School difficulties and role of social, material, behavioural, physical and mental resources among multi-cultural students**

Kénora Chau\(^{1,2}\) Michèle Baumann\(^2\) Bernard Kabuth\(^1\) Nearkasen Chau\(^{3,4}\)

We have made the changes (in red in the text). We have better describe our outcome variables Low-school performance and quitting school thinking in the methods section. We have completed the titles of the tables.

This manuscript has been seen and approved by all authors, which have been personally and actively involved in substantive work leading to this article, and will hold themselves jointly and individually responsible for its content. The manuscript is not simultaneously under consideration for publication in another journal and the material has not been published elsewhere. All relevant ethical safeguards have been met in relation to subjects’ protection. We have no conflict of interest to declare.

Looking forward to hearing from the Editorial Board,
On behalf of the authors,
Sincerely yours,

Prof Michèle BAUMANN
University of Luxembourg
Research Unit INSIDE
Route de Diekirch BP 2